Author's response to reviews

Title: Technology assessment and resource allocation for predictive genetic testing: A study of the perspectives of Canadian genetic health care providers

Authors:

Alethea Adair (aadair@ualberta.ca)
Robyn Hyde-Lay (rhyde-lay@genomealberta.ca)
Edna Einsiedel (einsiede@ucalgary.ca)
Timothy Caulfield (tcaulfld@law.ualberta.ca)

Version: 5 Date: 1 June 2009

Author's response to reviews: see over